Author's response to reviews

Title: The influence of regional basic science campuses on medical students' choice of specialty and practice location: a historical cohort study

Authors:

James J Brokaw (jbrokaw@iupui.edu)
Christina A Mandzuk (christina.mandzuk@gmail.com)
Michael E Wade (mwade321@comcast.net)
Dennis W Deal (ddeal@iupui.edu)
Mary T Johnson (johnsomt@iupui.edu)
Gary W White (gwhite@usi.edu)
Jeffrey S Wilson (jeswilso@iupui.edu)
Terrell W Zollinger (tzolling@iupui.edu)

Version: 3 Date: 30 May 2009

Author's response to reviews: see over
May 30, 2009

Editorial Production Team
BMC Medical Education

RE: Manuscript 1187888115252310

Dear Editorial Production Team:

I am submitting a revised manuscript entitled, “The Influence of Regional Basic Science Campuses on Medical Students’ Choice of Specialty and Practice Location: A Historical Cohort Study” by Brokaw et al. I have made all of the major and minor formatting changes requested, as described below.

Major Revisions
- Missing departmental information has been added for two of the authors (MTJ and GWW).
- The “About the authors” section is complete as currently written.
- Colors have been removed from all tables.
- Figure numbers have been removed from all image files.
- The “Submitted as separate Microsoft Word files” section has been formatted as instructed. As per instructions on the BMC website, Tables 2 and 3 were submitted as additional files because they are in landscape format. However, I hope they can be incorporated into the manuscript along with Table 1 and the figures.

Minor Revisions
- All extraneous text has been removed from the title page.
- All figures have been cropped to minimize white space.

No other changes have been made to the manuscript. I hope the paper is now suitable to publish. Please contact me if you require any further changes.

Thank you for your assistance.

Sincerely,

James J. Brokaw, PhD, MPH
Associate Dean for Medical Student Affairs
Indiana University School of Medicine
Phone: 317-278-2891
Email: jbrokaw@iupui.edu